Author's response to reviews

Title: Recurrent wheezing is associated with intestinal protozoan infections in Warao Amerindian children in Venezuela: a cross-sectional survey

Authors:

Marcella M.A. Overeem (marcellaovereem@hotmail.com)
Lilly M. Verhagen (lillyverhagen@hotmail.com)
Peter W.M. Hermans (p.hermans@ckz.umcn.nl)
Berenice del Nogal (berenicedelnogal@yahoo.es)
Adriana Márquez Sánchez (marquezadriana634@gmail.com)
Natacha Martínez Acevedo (natacham30@hotmail.com)
Rosalicia Ramírez Murga (rrosaliciam@gmail.com)
Jeroen Roelfsema (jeroen.roelfsema@rivm.nl)
Elena Pinelli (elena.pinelli.ortiz@rivm.nl)
Jacobus de Waard (jacobusdeward@gmail.com)

Version: 2 Date: 2 March 2014

Author's response to reviews: see over
Dear Prof. Harris,

It is a great pleasure to submit to BMC Infectious Diseases our revised manuscript entitled ‘Recurrent wheezing is associated with intestinal protozoan infections in Warao Amerindian children in Venezuela: a cross-sectional survey’.

We thank the reviewers for their constructive comments and suggestions. All suggestions made by the reviewers have been extensively discussed and, where possible, implemented in the revised manuscript. The explanatory details are listed in the enclosed document.

Thanks to the highly relevant suggestions of the reviewers, our revised manuscript has improved significantly. We are confident that the current manuscript meets the quality criteria of BMC Infectious Diseases, and that the subject will be appreciated by the Journal’s readership.

All authors have contributed significantly to the work. This manuscript has not been published and is not being considered for publication elsewhere. All authors have read and approved the manuscript. The order of authors listed in the manuscript has been approved by all authors.

We hope hearing from you at your earliest convenience.

Sincerely yours,

Marcella M.A. Overeem, MSc, corresponding author
Laboratory of Pediatric Infectious Diseases
Radboud University Nijmegen Medical Centre
PO Box 9101 (internal post 224)
6500 HB Nijmegen
The Netherlands
T +31-24-3666406 (office)
T +31-24-3666407 (secr.)
F +31-24-3666352
e-mail: marcellaovereem@hotmail.com